

Format for ANSWERING REVIEWERS

May 29, 2015

Dear Editor,



Please find enclosed the edited manuscript in Word format (file name: 2429-review.doc).

Title: " Malignant Peripheral Nerve Sheath Tumour Of Proximal Third Tibia Treated By En-Bloc Resection"

Author:

ArunkumarRao, Sachin B Ingle ,PawanRajurkar, VishavGoyal, Nikhil Dokrimare

Name of Journal: *World Journal of Clinical Oncology*

ESPS Manuscript NO: 17025

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

- (1) Grammatical errors and Spelling mistakes are corrected
- (2) Negative history of radiation exposure and Neurofibromatosis is mentioned in the case report
- 3) Immunohistochemistry figure is added
- 4) Core tip and PMID for the references are added

.....

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Clinical Oncology*

A handwritten signature in blue ink, appearing to read 'S B Ingle', written over the word 'Sincerely'.

Sincerely

Prof S B Ingle

MIMSR Medical College, Latur

Maharashtra, India